\documentclass[10pt,letterpaper, final]{report} % \documentclass{} required in all files

\usepackage{../preamble/myWEPS} % This is now our single preamble file inclusion (it assumes a “.sty” file extension)

\begin{document}

The Run Summary report (\cref{fig:WEPS_Run_Summary_Report_pg1})
will automatically display at the conclusion of a WEPS simulation run.
If the Run Summary report has been closed, the user can re-display the 
Run Summary for the most recent WEPS Run by selecting the
\scalerel*{\includegraphics{images/icons/ReportR.png}}{\strut} button
on the main WEPS screen button bar or a previous WEPS Run
by clicking the “ViewOutput” menu and selecting the “Previous Run”
and then the desired “Run” option from the WEPS main screen menu bar. 

\begin{figure}[H]
\centering
	\includegraphics[scale=2.0]{images/Reports/WEPS_Run_Summary_Report.jpg}
\caption{First page of WEPS Run Summary Report.}
\label{fig:WEPS_Run_Summary_Report_pg1}
\end{figure}

The Run Summary contains the header information, \hyperlink{Report_Header}{\textbf{previously mentioned here}},
along with the following additional information:

\begin{itemize}[noitemsep]
\item{Graphical representation of the field site}
\item{Dimensions, orientation, area and elevation of the field site}
%\item{What “simulation mode” and number of “cycles” were used for the WEPS Run}
\item{What “simulation mode” was used for the WEPS Run}
\item{Soil Loss Tolerance (T) value for the specified soil}
\item{Site location, e.g. country, state and county, latitude and longitude coordinates, etc.}
\item{Name of Cligen station used in WEPS Run}
\item{Name of Windgen station, or if interpolated, the “lat/long” coordinates, used by Windgen}
\end{itemize}

Here is an example of the Run Summary \textbf{Location Site Information} section as shown in
\cref{fig:Location_Site_Information}.

\begin{figure}[H]
\centering
	\includegraphics[scale=2.0]{images/Reports/Location_Site_Information.jpg}
\caption{WEPS Run Summary Report's "Location Site Information" section.}
\label{fig:Location_Site_Information}
\end{figure}


The WEPS Run Summary report consists of the following report sections which provide:

\begin{itemize}[noitemsep]
\item{Wind erosion soil loss by rotation year and annual average}
\item{Average erosion rates by Crop Interval periods}
\item{Barrier type and properties (when present on the field borders)}
\item{Summarized harvested crop information}
\item{SCI (Soil Conditioning Index) summary}
\item{Rotation Cover/Standing Silhouette/STIR summary for the management rotation}
\item{Crop Interval period STIR Energy summary}
\item{WEPS Run user notes}
\item{WEPS Run user notes (editable section)}
\end{itemize}

\subsection{Erosion (Summary Report Section)} \label{subsec:Erosion (Summary Report Section}
This section of the Run Summary reports the wind erosion results from the simulation by specific periods
in a tabular section titled \textbf{Erosion} (\cref{fig:WEPS_Run_Summary_Report_pg1} and \cref{fig:WEPS_Run_Summary_Erosion_report}).
The values are reported in units of mass per unit area, e.g. for SI (kg/m$^2$) and English (t/ac).
The columns of this report section consist of:

\begin{itemize}[label={}]

\item{Column 1, labeled \textbf{Period}.
This column consists of the individual rotation years
within the management/crop rotation cycle (Rot. Year: $\#$)
and the complete management/crop rotation cycle (Ave. Annual).}

\item{Column 2, labeled \textbf{Crop/Residue}.
This column consists of the crops harvested during the specified rotation years.}

\item{Column 3, labeled \textbf{Gross Loss}.
This column contains the average erosion within the field
(i.e., removal of soil with no deposition taken into account).}

\item{Columns 4-7 labeled \textbf{Net Soil Loss From Field}.
These columns consists of the average total net soil losses
from the field (net losses are gross losses minus deposition within the field).
These values are broken down and reported by the size class of the particulates.}

\item{Column 4, labeled \textbf{Total}.
This column contains the total average net soil loss from the field for all size classes.}

\item{Column 5, labeled \textbf{Creep/Salt.}.
This column consists of the average net soil loss from the field for creep and saltation size particles ($\ge$100µ)}.

\item{Column 6, labeled \textbf{Suspen.}.
This column contains the average net soil loss from the field for suspension size particles (${<}$100µ)}.

\item{Column 7, labeled \textbf{PM10}.
this column holds the average net soil loss from the field for the fraction of suspension size particles less than 10µ in size.}
\end{itemize}

\begin{figure}[H]
\centering
	\includegraphics[scale=2.0]{images/Reports/run_summary_erosion_report.jpg}
\caption{WEPS Run Summary Erosion report.}
\label{fig:WEPS_Run_Summary_Erosion_report}
\end{figure}

Deposition occurs when the wind speed drops as it travels across the field
or the surface conditions become less erodible in a downwind part of the field
(i.e., transport capacity of the wind exceeds transport capacity for the surface conditions).
In \WEPSver, deposition only occurs with a drop in wind speed across the field
due to a barrier present on the downwind side of the field or surface conditions
changing across the field because of the erosion process.
Therefore, since \WEPSver
only simulates an entire field as consisting of a single soil type
and only one sequence of management practices applied uniformly to that field,
it is not uncommon to have equal gross and net total soil loss when no field barriers are included
in the simulation\footnote{Later releases of WEPS will allow multiple soil types
and management sequences to be simulated on a field,
so the effects of soil type and management sequence changes across a field can be represented.
This will also allow more sophisticated practices
such as strip cropping to be properly represented in WEPS.}.

If an erosion event occurred, but values generated by the model are too small to be displayed
on the output table (i.e., ${<}$0.001 kg/m$^2$ when displaying SI units
or ${<}$0.1 t/ac for saltation/creep and ${<}$0.01 t/ac for PM10 when displaying English units),
then the amount is listed as “trace”.
If amounts exceed the range of the measured data used for calibrating WEPS,
the values are still reported, but are presented with a warning
(\cref{fig:WEPS_Run_Summary_Erosion_report_displaying_warnings}).
The values exceeding the specified value (i.e., ${>}$ 300 kg/m$^2$ or ${>}$100 t/ac)
are noted with a yellow symbol
\scalerel*{\includegraphics{images/icons/warning.png}}{\strut}
adjacent to those values.
In these cases, erosion amounts are so large that they are generally unacceptable levels
for conservation planning and erosion control purposes.


\begin{figure}[H]
\centering
	\includegraphics[scale=2.0]{images/Reports/Erosion_warning_msg.jpg}
\caption{WEPS Run Summary Erosion report displaying warnings.}
\label{fig:WEPS_Run_Summary_Erosion_report_displaying_warnings}
\end{figure}

\subsection{Crop Interval Erosion (Summary Report Section)} \label{subsec:Crop Interval Erosion (Summary Report Section)}

This section of the Run Summary (\cref{fig:WEPS_Run_Summary_Crop_Interval_report})
reports the wind erosion results from the simulation by crop interval,
which is defined as the time from the previous crop’s termination (final harvest or crop kill process)
until the termination of this crop (final harvest or crop kill process).
The soil loss values (last 5 columns) are reported exactly as they are for the \textbf{Erosion}
sub-report previously described. The first three columns consist of:

\begin{itemize}[label={}]
\item{Column 1, labeled \textbf{Date Range}. Column consists of the start and ending dates
for each crop interval.}

\item{Column 2, labeled \textbf{Days}. Column consists of the crop interval period length in days.}

\item{Column 3, labeled \textbf{Crop}. Column consists of the crops defining each “crop interval”.}

\item{Columns 4-7, labeled \textbf{Net Soil Loss From Field}.
These columns consists of the average total net soil losses
from the field (net losses are gross losses minus deposition within the field).
These values are broken down and reported by the size class of the particulates.}

\item{Column 4, labeled \textbf{Total}.
This column contains the total average net soil loss from the field for all size classes.}

\item{Column 5, labeled \textbf{Creep/Salt.}.
This column consists of the average net soil loss from the field for creep and saltation size particles ($\ge$100µ)}.

\item{Column 6, labeled \textbf{Suspen.}.
This column contains the average net soil loss from the field for suspension size particles (${<}$100µ)}.

\item{Column 7, labeled \textbf{PM10}.
This column holds the average net soil loss from the field for the fraction of suspension size particles less than 10µ in size.}
\end{itemize}

\begin{figure}[H]
\centering
	\includegraphics[scale=2.0]{images/Reports/crop_interval_subreport.jpg}
\caption{WEPS Run Summary Crop Interval report.}
\label{fig:WEPS_Run_Summary_Crop_Interval_report}
\end{figure}

\subsection{Harvests (Summary Report Section)} \label{subsec:Harvests (Summary Report Section)}
This section of the Run Summary (\cref{fig:WEPS_Run_Summary_Harvests_report})
provides a summarization of the average yield and after-harvest residue amounts
for each harvested crop grown in the management/crop rotation scenario.
The information is presented in tabular form where the columns consist of:

\begin{itemize}[label={}]
\item{Column 1, labeled \textbf{Date}. This column consists of the harvest date for the specified crop.}
\item{Column 2, labeled \textbf{Crop}. This column consists of the crops that are harvested.}
\item{Column 3, labeled \textbf{Residue}.
This column consists of the average after harvest crop residue left on the field and 
is reported in kg/m$^2$ when SI units are specified and lbs/ac when English units are used.}
\item{Column 4, labeled \textbf{Harvest Yield}.
This column consists of the average yield produced by the crop
and is reported in kg/m$^2$ when SI units are specified.
When English units are used, the units displayed are obtained from the crop record,
e.g. ton/ac, bu/ac, etc.}
\item{Column 5, labeled \textbf{Yield $\%$ Moisture}.
This column consists of the percent water content value specified in the 
crop record and is used for reporting the yield.}
\end{itemize}

\begin{figure}[H]
\centering
	\includegraphics[scale=2.0]{images/Reports/run_summary_harvests.jpg}
\caption{WEPS Run Summary Harvests report.}
\label{fig:WEPS_Run_Summary_Harvests_report}
\end{figure}

\subsection{SCI Summary (Summary Report Section)} \label{subsec:SCI Summary (Summary Report Section)}
This section of the Run Summary (\cref{fig:WEPS_Run_Summary_SCI_report})
provides a summary of the SCI (Soil Condition Index),
Energy Calculator, Average STIR, and other inputs required for computing the SCI,
including the SCI subfactors.
The SCI is an index developed by NRCS to reflect trends in soil organic matter,
which are assumed to be an indicator of soil quality trends.
A negative value generally implies that the soil organic matter is decreasing
and thus soil quality would be degrading over time.
A positive value implies that soil organic matter is increasing
and thus soil quality is generally improving over time.
A value of zero or near zero implies that soil organic matter is not changing over time
and thus the soil quality is sustainable.
The values presented in this summary are:

\begin{itemize}[label={}]
\item{\textbf{Soil Condition Index} - Index developed and used by NRCS to reflect the relative quality of a soil due to the management/crop rotation practices being applied.}
\item{\textbf{Energy Calculator} - Average annual energy per acre expended for operations specified in the management/crop rotation.}
\item{\textbf{Average Annual STIR} - Computed for the management/crop rotation sequence from the listed operations}
\item{\textbf{Wind Erosion Soil Loss} - An input required for SCI calculation. Obtained from the WEP simulation run.}
\item{\textbf{Water Erosion Loss} - An input required for SCI calculation. User is expected to specify the value on the main WEPS screen prior to the WEPS simulation being executed.}
\end{itemize}

{\leftskip=0.5cm\relax
\textbf{SCI Subfactors} presented in this sub-report are:
\par}

\begin{itemize}[label={}]
\item{\textbf{OM} - It is the organic material or biomass subfactor.
This component accounts for the effect of biomass returned to the soil,
including material from plant or animal sources,
and material either imported to the site or grown and retained on the site.
It uses the initial soil surface/layer properties as provided at the beginning of a WEPS simulation.}
\item{\textbf{FO} - It is the field operations subfactor.
This component accounts for the effect of field operations that stimulate organic matter breakdown. 
Tillage, planting, fertilizer application, spraying, harvesting,
and other operations crush and shatter plant residues and aerate or compact the soil.
These effects increase the rate of residue decomposition
and affect the placement of organic material in the soil profile.}
\item{\textbf{ER} - It is the erosion subfactor.
This component accounts for the effect of removal and sorting of surface soil organic matter
by sheet, rill, or wind erosion processes as predicted by water and wind erosion models.
It does not account for the effects of concentrated flow erosion,
such as ephemeral or classic gullies.}
\end{itemize}

\begin{figure}[H]
\centering
	\includegraphics[scale=2.0]{images/Reports/Run_summary_SCI_summary.jpg}
\caption{WEPS Run Summary SCI report.}
\label{fig:WEPS_Run_Summary_SCI_report}
\end{figure}

\subsection{Rotation Cover/Standing Silhouette/STIR (Summary Report Section)} \label{subsec:Rotation_Cover_Standing_Silhouette_STIR (Summary Report Section)}

This section of the Run Summary
(\cref{fig:WEPS_Run_Summary_Rotation_Cover_Standing_Silhouette_STIR_report})
provides a summarization
of the surface residue cover and effective standing silhouette area
following each management operation
as well as the STIR (Soil Tillage Intensity Rating) value for the operations listed
in the management/crop rotation file.
The flat and standing residue information provides the WEPS user with an
indication of the amount of wind erosion protection provided by the residue on those dates
following the listed operations.
The STIR value is an indication of the overall “intensity” of the tillage applied
for each specific operation in a given management/crop rotation scenario.
The total rotation and average yearly rotation STIR values are also provided.

\begin{figure}[H]
\centering
	\includegraphics[scale=2.0]{images/Reports/WEPS_Run_Summary_Rotation_STIR_Energy_subreport.jpg}
\caption{WEPS Run Summary Rotation Cover/Standing Silhouette/STIR report.}
\label{fig:WEPS_Run_Summary_Rotation_Cover_Standing_Silhouette_STIR_report}
\end{figure}

The Rotation Stir Energy table columns are:
\begin{itemize}[label={}]
\item{Column 1, labeled \textbf{Date}.
This column consists of operation date for the specified operation.}
\item{Column 2, labeled \textbf{Operation}.
This column consists of the operation used on the specified date.}
\item{Column 3, labeled \textbf{Crop (planted)}.
This column consists of crop planted by the specified operation.}
\item{Column 4, labeled \textbf{Surface Cover}.
This column consists of the residue surface cover remaining
following the specified operation.}
\item{Column 5, labeled \textbf{Eff. Standing Silhouette}.
This column consists of the remaining standing residue, reported as
effective standing silhouette area (ft$^{2}$/ft$^{2}$ in English units and m$^{2}$/m$^{2}$ in SI units)
following the specified operation.} 
\item{Column 6, labeled \textbf{STIR}.
This column consists of the computed STIR value for the specified operation.}
\end{itemize}

The last two rows display the full management rotation cycle's STIR value and the average annual STIR value.

\subsection{Crop Interval Stir Energy (Summary Report Section)} \label{subsec:Crop Interval Stir Energy (Summary Report Section)}

This section of the Run Summary
(\cref{fig:WEPS_Run_Summary_Rotation_Cover_Standing_Silhouette_STIR_report})
provides a summarization
of the STIR (Soil Tillage Intensity Rating), Energy and Energy Costs
associated with each specific crop interval.
The STIR value is an indication of the overall "intensity"
of the tillage applied during each specific crop 
interval in a given management/crop rotation scenario.

\begin{figure}[H]
\centering
	\includegraphics[scale=2.0]{images/Reports/Run_summary_crop_interval_STIR_energy.jpg}
\caption{WEPS Run Summary Crop Interval Stir Energy report.}
\label{fig:WEPS_Run_Summary_Crop_Interval_Stir_Energy_report}
\end{figure}

The Crop Interval Stir Energy table columns are:

\begin{itemize}[label={}]
\item{Column 1, labeled \textbf{Date Range}.
This column consists of the beginning and ending dates for the specified crop interval.}
\item{Column 2, labeled \textbf{Days}.
This column consists of the number of days for the specified crop interval.}
\item{Column 3, labeled \textbf{Crop}.
This column consists of the crop grown during the specified crop interval.}
\item{Column 4, labeled \textbf{Stir}.
This column consists of the computed Stir value for the specified crop interval.}
\item{Column 5, labeled \textbf{Energy}.
This column consists of the total computed energy consumed per unit area (kJ/ha for SI units and diesel (gal)/ac for English units) for the specified crop interval.}
\item{Column 6, labeled \textbf{Cost}.
This column consists of the total computed cost in dollars per unit area (US dollars/ha for SI units and US dollars/ac for English units) for the specified crop interval.}
\end{itemize}

\subsection{Notes (Summary Report Section)} \label{subsec:Notes (Summary Report Section)}

This \textbf{Notes} section of the Run Summary
(\cref{fig:WEPS_Run_Notes_listed_in_Run_Summary})
consists of any text entered prior to the simulation
on the main WEPS screen’s \textbf{Notes} section (\cref{fig:WEPS_main_screen}).
Any notes entered on the main WEPS screen for this run are reproduced here.

\begin{figure}[H]
\centering
	\includegraphics[scale=2.0]{images/Reports/Run_summary_notes.jpg}
\caption{WEPS Run Notes listed in Run Summary.}
\label{fig:WEPS_Run_Notes_listed_in_Run_Summary}
\end{figure}

\subsection{Editable Notes (Summary Report Section)} \label{subsec:Notes (Editable Section)}

This editable \textbf{Notes} section consists of two parts, the “Notes" tab and the \hl{“Warnings”} tab
(\cref{fig:WEPS_Run_Notes_listed_in_Run_Summary_Editable})
They can be edited or added to, if desired, and saved via the appropriate button
{\scalerel*{\includegraphics{images/icons/savenotes.png}}{\strut} at the top of the Notes form.
Any warnings generated during the simulation are listed under the “Warnings” tab.


\begin{figure}[H]
\centering
	\includegraphics[scale=2.0]{images/Reports/Run_summary_notes_editable.jpg}
\caption{WEPS Run Notes listed in Run Summary (editable section).}
\label{fig:WEPS_Run_Notes_listed_in_Run_Summary_Editable}
\end{figure}

\end{document}